ECONSTOR

Make Your Publications Visible.

A Service of



Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Kieren, Pascal; König-Kersting, Christian; Schmidt, Robert J.; Trautmann, Stefan T.; Theurich, Franziska

Working Paper First-order and higher-order inflation expectations: Evidence about households and firms

Working Papers in Economics and Statistics, No. 2023-10

Provided in Cooperation with: Institute of Public Finance, University of Innsbruck

Suggested Citation: Kieren, Pascal; König-Kersting, Christian; Schmidt, Robert J.; Trautmann, Stefan T.; Theurich, Franziska (2023) : First-order and higher-order inflation expectations: Evidence about households and firms, Working Papers in Economics and Statistics, No. 2023-10, University of Innsbruck, Research Platform Empirical and Experimental Economics (eeecon), Innsbruck

This Version is available at: https://hdl.handle.net/10419/278557

Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



WWW.ECONSTOR.EU



Faculty of Economics and Statistics



First-Order and Higher-Order Inflation Expectations: Evidence about Households and Firms

Pascal Kieren, Christian König-Kersting, Robert Schmidt, Stefan Trautmann, Franziska Heinicke

Working Papers in Economics and Statistics

2023-10



University of Innsbruck Working Papers in Economics and Statistics

The series is jointly edited and published by

- Department of Banking and Finance
- Department of Economics
- Department of Public Finance
- Department of Statistics

Contact address of the editor: Faculty of Economics and Statistics University of Innsbruck Universitaetsstrasse 15 A-6020 Innsbruck Austria Tel: + 43 512 507 96136 E-mail: Dean-EconStat@uibk.ac.at

The most recent version of all working papers can be downloaded at https://www.uibk.ac.at/fakultaeten/volkswirtschaft_und_statistik/forschung/wopec/

For a list of recent papers see the backpages of this paper.

First-Order and Higher-Order Inflation Expectations: Evidence about Households and Firms

Pascal Kieren¹, Christian König-Kersting², Robert Schmidt^{3*}, Stefan Trautmann^{1,4}, Franziska Heinicke⁵

¹ Heidelberg University, Germany
 ² University of Innsbruck, Austria
 ³ Deutsche Bundesbank, Germany
 ⁴ Tilburg University, The Netherlands
 ⁵ University of Mannheim, Germany

5 July 2023

Abstract: We study first-order and higher-order inflation expectations of German households and firms elicited from surveys. The data allows to shed light on the relation between different orders of beliefs, and to derivate implications for noisy-information models with infinite regress. Moreover, since the elicited data is identical for households and firms, it also allows studying whether the relation between first-order and higher-order beliefs differs between the two samples. While we find that this relation is mostly identical between households and firms in our data, we identify differences to previously elicited data in the literature. We discuss potential sources for these differences and their theoretical implications.

Highlights:

- We study first-order and higher-order inflation expectations of German households and firms
- The relation between first- and higher-order expectations is identical for both groups
- Implications for noisy-information models with infinite regress are discussed

Keywords:Inflation expectations, higher-order beliefs, noisy-information models, surveysJEL:D84, E31, G17

^{*}Corresponding author: Robert Schmidt, Wilhem-Epstein-Str. 14, 60431 Frankfurt am Main, Germany; robert.schmidt1@bundesbank.de. We thank the seminar audience in Heidelberg for very valuable comments and suggestions. Financial support from the Austrian Science Fund FWF through SFB F63 is gratefully acknowledged. Funding by the German Federal Ministry of Education and Research (BMBF) and the Baden-Württemberg Ministry of Science as part of Germany's Excellence Strategy is gratefully acknowledged. This paper represents the authors' personal opinions and does not necessarily reflect the views of the Deutsche Bundesbank.

1. Introduction

Individuals' expectations about the future value of goods and services play an important role in all economic settings. Yet, given the dynamic nature of interactions in markets, market participants do not only need to estimate fundamental values, but also others' expectations regarding these values. These expectations can vary substantially across individuals or entities, because of different information sets or because of alternative views about the development of the economy.

Higher order expectations, i.e., the anticipation of the expectations of others, are therefore an important strategic component central to many economic scenarios, and have become increasingly emphasized and studied in macroeconomic and finance contexts. For instance, Bacchetta and Van Wincoop (2008) demonstrate that the difference between higher-order and own expectations is important for the link between current asset prices and investors' expectations about future assets' payoffs. Woodford (2002) shows how noisy private information can lead firms to change their prices very gradually because of the slow-moving higher-order beliefs about the actions of other firms. Nimark (2008) highlights the role of higher-order inflation expectations in pricing decisions for generating inflation inertia, i.e. the fact that aggregate inflation responds only gradually to shocks while price changes for goods are quite large. More recent work has emphasized that incorporating higher-order uncertainty in a New Keynesian model reduces the effectiveness of forward guidance which helps to resolve the "forward guidance" puzzle (e.g., Angeletos and Lian 2018; Farhi and Werning 2019; Gabaix 2020). Considering how important higher-order expectations of macroeconomic variables are for monetary policy, surprisingly few studies have explored them empirically until recently – presumably due to data constraints.

This study aims to further our understanding of higher-order macroeconomic expectations by utilizing two large representative surveys of German firms' and households' inflation expectations. The surveys ask households and managers not only about their own expectations (i.e., their first-order expectations), but also what they believe others expect (i.e., their higher-order expectations). Jointly, these surveys allow us to uncover the relationships between first-order and higher-order macroeconomic beliefs that can be used to discipline and test models of higher-order expectations. Using the inflation expectations of firm managers from New Zealand, Coibion et al. (2021, henceforth CGKR) were the first to study these relationships and formulate their results as a set of stylized facts. We test the robustness of these stylized facts in a much larger sample from a much

larger economy, and for a different time frame. Additionally, we provide the first empirical analysis of the relation between first- and higher-order inflation expectations for households. This allows us to document commonalities in how firms and households form higher-order expectations, which helps to guide the empirical search for the determinants and formation of inflation expectations and how these expectations affect financial decisions.

Our analysis is based on data from the Deutsche Bundesbank, in particular the Bundesbank Online Panel Households and the Bundesbank Online Panel Firms. Both panels put strong emphasis on inflation expectations and regularly elicit the one-year-ahead inflation rate via point and via probabilistic predictions. In addition to the regular first-order expectations questions, we added the corresponding questions regarding higher-order expectations for households (March 2021 wave) and for firms (Q4/October to December 2021 wave). These additional modules allow us to establish a number of stylized facts about the relation of first- and higher-order inflation expectations in our unique data sets, which can be summarized as follows.

First, we show that the average higher-order forecast of inflation is almost identical to the average first-order forecast of inflation for both households and firms. In general, our data show a positive correlation between first-order and higher-order beliefs for households and firms. However, the correlation is not perfect and is more pronounced for firms than for households. In other words, firm managers display a higher tendency to believe that competitors have expectations about the future rate of inflation similar to their own. Second, we show that the cross-sectional standard deviation of higher-order beliefs (disagreement) is substantial in our sample and that the disagreement in first-order beliefs is on average smaller than the disagreement in higher-order beliefs about inflation. This finding holds for both firms and households. It suggests that although managers and households disagree about how inflation will develop over the next year, there is even more disagreement about what they think others are predicting inflation to be. Third, the average uncertainty (i.e., the standard deviation of individuals' probabilities across inflation bins) around households' and firms' own inflation forecast is significantly lower than their uncertainty around higher-order beliefs about inflation. This suggests that uncertainty accumulates as individuals extrapolate from their beliefs to what others might believe. Finally, when comparing uncertainty and disagreement, our results consistently show that the average degree of uncertainty is lower than average disagreement, i.e. the cross-sectional dispersion. This finding holds for firstand higher-order expectations as well as for firms and households. As we will show below, our results deviate in important ways from the stylized facts put forward by CGKR.

Our article contributes to several strands of literature. First, we contribute to the literature using survey data to understand households' and firms' macroeconomic expectations. For inflation, an extensive literature considers expectations of households (e.g., Bruine de Bruin et al., 2010; D'Acunto et al., 2021a), firms (e.g., Coibion et al., 2020a), investors (e.g., Coibion et al., 2020b), and even central bankers (e.g., Mankiw et al., 2003). However, evidence on higher-order expectations is still scarce and has thus far only been investigated for firm managers. We contribute a set of novel insights into both firms' and households' higher-order inflation expectations based on large representative samples. The established relationships between first- and higher-order expectations are remarkably consistent between our firm and household samples, but differ in important aspects from what CGKR report. Most notably, while CGKR find that both uncertainty and disagreement are higher in first-order than in second-order expectations, we find the opposite. Understanding how heterogeneity in own beliefs compares to the heterogeneity in what others belief is important as it can facilitate sharper tests of macroeconomic models for which subjective beliefs are a driver of economic activity. Mankiw and Reis (2002) as well as Hellwig and Venkateswaran (2009) show that with heterogenous information, firms face uncertainty not only about aggregate fundamentals but also about the pricing decision of other firms because they can no longer be sure that other firms have the same information. Our results highlight that another source of uncertainty is how agents extrapolate from their own information to what other people believe, even if they share the same information. In essence, our findings imply that uncertainty accrues as individuals move from their own expectations to what other people believe.

Second, our findings contribute to the literature on how agents form expectations in strategic environments. Following early work by Morris and Shin (1998), a growing body of theoretical work emphasizes the potential importance of incomplete information and higher-order thinking for optimal policy and understanding economic dynamics (e.g., Angeletos and La'O, 2009). Establishing the intricate relationship between first- and second-order expectations provides empirical benchmarks for building and testing theories of expectation formation. Applied to inflation expectations, CGKR consider multiple extensions to the static noisy information model of Morris and Shin (2002) to reconcile their stylized facts regarding higher-order expectations with theoretical predictions. In a similar spirit, we consider an extension to the static noisy-information model which introduces heterogeneity in prior beliefs as proposed by Patton and Timmermann (2010). On the one hand, this extension allows us to establish conditions which help to explain the discrepancy between our stylized facts and CGKR's. On the other hand, this exercise demonstrates

that the introduction of long-run priors is a promising candidate to further our understanding of how individuals form higher-order expectations. Consistent with this notion, we present evidence that supports the existence of heterogenous long-run priors as suggested by Patton and Timmermann (2010).

Our paper proceeds as follows. Section 2 describes the implementation of the surveys and how the first- and higher-order inflation expectations were measured in the respective modules. Section 3 presents our main findings. Section 4 considers the implications for noisy-information models in the light of our empirical rejection of the stylized facts established by CGKR. Section 5 concludes the paper.

2. Data and Survey Design

2.1. The Bundesbank Online Panels for Households and Firms

We use microdata from the Bundesbank Online Panel Households (BOP-HH) and the Bundesbank Online Panel Firms (BOP-F). The BOP-HH (BOP-F) is a regular monthly (quarterly) survey of German households (firms) and was established in April 2019 (June 2020), with the first waves being considered as pilot waves before entering a steady state. The waves in each survey contain both a panel component and a refresher sample, and they consist of a set of core questions as well as additional modules for the investigation of specific topics. The topics concern various expectations and assessments in the economic, political and social domain. In addition, households (firms) answer various questions regarding socio-demographics or household characteristics (firm characteristics). The panels reflect a sample representative for the German population and the German firm sector, respectively.

2.2 Measurement of Inflation Expectations

Both panels put strong emphasis on inflation expectations and regularly elicit the one-year-ahead inflation rate via both point and probabilistic predictions. In addition to conventional first-order expectations, we added the corresponding questions regarding higher-order expectations for households (March 2021 wave) and for firms (Q4/October to December 2021 wave). For the respective periods, the data set contains about 2,300 observations for first-order and higher-order inflation expectations of households and about 2,700 observations for first-order and higher-order

inflation expectations of firms.¹ In both cases, the data set contains expectations in the form of point expectations (decimal number with one decimal place) and probabilistic expectations where subjects distribute probability mass over different inflation intervals (Manski, 2004).² Finally, the data also contain a series of waves where three-year, five-year and ten-year ahead point inflation expectations are elicited from households and firms. In Section 3, we concentrate on expectations regarding the one-year ahead inflation rate. In Section 4, we additionally include data on long-term inflation expectations in order explain potential differences between our results and previously found patterns in the literature. We also use the combined data on inflation expectations on different time horizons to inform noisy-information models with infinite regress.

3. Results

We structure the results as follows. In section 3.1, we analyze the relation between the most important moments of first-order and higher-order inflation expectations. This analysis follows the stylized facts identified by CGKR. We test whether these facts hold in our household and firm samples. In section 3.2, we analyze the prediction error, i.e., the inaccuracy of higher-order expectations, and its relation with individual-level uncertainty in expectations. Section 3.3. studies the role of household and firm characteristics.

¹ Table A1 in Appendix A provides an overview about the distribution of several household characteristics (gender, age, income, education) and firm characteristics (number of employees, turnover, firm sector, firm region).

² Appendix B contains an overview of the exact wording of all the questions regarding first-order and higher-order inflation expectations we use in our analyses.

Panel A. Mean inflation expectations (in %) A1. Households



Notes: FO = first-order, HO = higher-order. The data from Panel A and Panel B contain both the point and the probabilistic expectation. Data on point expectations is truncated below -12 and above +12. The implied mean derived from the probabilistic estimation is calculated using a value of -16 and 16, respectively, for the outer bins. Panel B represents the between-subject disagreement measured as the standard deviation of inflation expectations. Uncertainty in Panel C is measured as the standard deviation of the probabilistic inflation expectations.

3.1. First-Order and Higher-Order Inflation Expectations

Figure 1 contains descriptive results of all waves. It shows the available panel history of mean firstorder inflation expectations, disagreement (i.e., the cross-sectional standard deviation), and uncertainty (i.e., the standard deviation of the probabilistic inflation expectations) of both households and firms. In addition, the figure shows the corresponding values regarding higherorder expectations for the respective waves, i.e., for households in March 2021 and for firms in Q4 2021. Table 1 provides the main findings.

	Obs.	Mean	Disagreement	Uncertainty	Correlation
	(1)	(2)	(3)	(4)	(5)
Households:					· ·
Point Expectations					
First-order	2,306	2.65	2.08	-	1.00
Higher- order	2,306	2.75	2.21		0.59
Difference (<i>p</i> -value for equality		-0.10	-0.13	-	
of moment)		(0.02)	(<0.01)		
Households:					
Probabilistic Expectations					
First-order	2,310	2.24	2.78	1.68	1.00
Higher- order	2,310	2.20	2.78	2.16	0.67
Difference (<i>p</i> -value for equality		0.04	0.00	-0.48	
of moment)		(0.39)	(0.99)	(<0.01)	
Firms:					
Point Expectations					
First-order	2,681	4.09	1.57	_	1.00
Higher- order	2,681	4.13	1.55		0.71
Difference (<i>p</i> -value for equality		-0.03	0.02	-	
of moment)		(0.16)	(0.60)		
Firms:					
Probabilistic Expectations					
First-order	2,718	3.51	2.54	1.22	1.00
Higher- order	2,718	3.37	2.64	1.36	0.68
Difference (<i>p</i> -value for equality		0.14	-0.10	-0.14	
of moment)		(0.01)	(0.04)	(<0.01)	

 Table 1. First-order and higher-order inflation expectations

Notes: The table reports basic moments of first-order and higher-order inflation expectations. Household data has been elicited in March 2021 and firm data between October and December 2021. In order to avoid extreme values, the data on point expectations is truncated below -12 and above +12. The implied mean from the probabilistic estimation in column (2) is calculated using a value of -16 and 16, respectively, for the outer bins. Disagreement in column (3) reports the cross-sectional standard deviation of mean inflation forecasts. Uncertainty in column (4) refers to the standard deviation of the reported probability distribution for future inflation using a value of -16 and 16, respectively, for the outer bins. Column (5) reports the correlation with first-order expectations. Tests for the equality of moments in column (2) and (4) are based on Wilcoxon signed-rank tests. Tests for the equality of disagreement in column (3) are based on Levene's test of homogeneity of variances.

3.1.1 Means and Correlations

For households, we find the first-order explicit point expectation to be slightly lower than the means derived from higher-order expectation (2.65 vs. 2.75, p = 0.02). The implied first-order and higher-order mean expectation derived from the probabilistic measure do not differ (2.24 vs. 2.20, p = 0.39). For firms, the explicit first-order and higher-order point expectations do not differ (4.09 vs. 4.13, p = 0.16), but the first-order implied mean expectation is slightly higher than the higher-order implied point expectation (3.51 vs. 3.37, p = 0.01). Taken together, first-order expectations are neither consistently higher nor lower than higher-order expectations; apparent differences are generally small in absolute terms and their economic relevance may be questioned. The data is consistent with CGKR, who likewise find that for firms average first-order inflation expectations do not differ systematically from average higher-order inflation expectations. Our data generalizes this finding to households and indicates that it is also insensitive to the elicitation method (point versus probabilistic mean expectation).³

After comparing first-order and higher-order expectations on the aggregate level, we analyze the coherence between first-order and higher-order inflation expectations on the individual level. Figure 2, Panel A, shows the distribution of individual-level differences between mean first-order and higher-order point expectations.⁴ For both households and firms, the difference between firstorder and higher-order point expectations is symmetrically distributed around zero, with zero being the modal difference on the individual level. However, the coherence between first-order and higher-order expectations is lower for households than for firms; the distribution of differences is less dispersed for firms and first-order and higher-order expectations are identical more often (28.3% for households and 39.1% for firms; p < 0.001, Mann-Whitney-U-Test). Panel B shows the analogous analysis for probabilistic expectations by comparing the distribution of withinrespondent differences in probabilities assigned to first- and higher-order beliefs. The data shows that most households and most firms do not assign identical weights to the same bins for first-order and higher-order expectations. Instead, they assign relatively more weight to the middle bins for first-order beliefs and relative more weight to the outer bins for higher-order beliefs. Specifically, the shapes of the red lines indicate that the probability distribution referring to first-order beliefs is more concentrated than the probability distribution referring to higher-order beliefs. Also, as in the

³ In Appendix C we provide some insights on the relationship between the point predictions and implied means of the respective probabilistic measures.

⁴ Figure D1 in Appendix D shows the distributions of first- and higher-order expectations separately.

case of point expectations, the coherence between first-order and higher-order expectations tends to be higher for firms than for households: First, the red line is flatter for firms than for households; second, while households assign identical weights to the respective bins in 17.0% of all cases, firms do so in 33.9% (p < 0.001, Mann-Whitney-U-Test).⁵

Result 1. CGKR Fact 1 is confirmed. In the aggregate, first-order and higher-order inflation expectations closely correspond for both households and for firms. However, on the individual level, there is a fair amount of incoherence between first-order and higher-order expectations. For both point and probabilistic beliefs, first-order and higher-order expectations diverge more often for households than for firms.

⁵ Note that subjects that exhibit the identical probability distributions for first-order and higher-order expectations fall into the white area in each bin in Panel B of Figure 2. Figure D2 in Appendix D contains analyses about the distribution of differences between first-order and higher-order inflation expectations by subsamples.

Figure 2. Distribution of differences between first-order and higher-order inflation expectations

Panel A. Point expectations: Difference between mean first-order and higher-order expectationsA1. HouseholdsA2. Firms



Panel B. Probabilistic expectations: Distribution of within-respondent differences in probabilities assigned to first- and higher-order beliefs



Notes: Panel A refers to households' and firms' point expectations. The figure shows the distribution of the differences between first-order and higher-order point estimation (i.e., first-order point expectation minus higher-order point expectation). Panel B refers to households' and firms' probabilistic expectations. The white area reflects the category zero in the legend. The red circles are the average difference (first-order minus higher-order expectation) in probability assigned to a specific bin. The sum of the red circles is 100% and the flatter the red line, the higher is the correspondence between first-order and higher-order inflations expectations.

Correlation analysis corroborates these findings. We find a positive correlation between firstorder and higher-order expectations across both samples and both elicitation methods. However, correlations are clearly lower than 1. For households, we find correlation coefficients of 0.59 and 0.67 for the point and the implied mean estimates, respectively. For firms, the corresponding values are 0.71 and 0.68. The correlation between first-order and higher-order point predictions is lower for households than for firms (0.59 vs. 0.71; p < 0.001), while the correlation between the probabilistic expectations does not differ between households and firms (0.67 vs. 0.68; p = 0.39).⁶

Our data is consistent with CGKR, who likewise find that first-order and higher-order inflation expectations are positively but not perfectly correlated, and the sizes of correlation coefficients are very similar to those found by CGKR. Our data generalizes their finding to households.

Result 2. CGKR Fact 2 is confirmed. The correlation between first-order and higher-order expectations is positive though not perfect, for both households and firms and for both point and probabilistic expectations. For point expectation, the correlation between first-order and higher-order expectations is higher for firms than for households.

3.1.2. Disagreement

For households, the cross-sectional disagreement is lower for first- than for higher-order expectations in the case of the point predictions (2.08 vs. 2.21; p < 0.01, Levene's test of homogeneity of variances), and identical for first- and higher-order expectations based on implied means from the probabilistic measure (2.78 vs. 2.78; p = 0.99, Levene's test of homogeneity of variances). For firms, the disagreement in point predictions is identical (1.57 vs. 1.55; p = 0.60, Levene's test of homogeneity of variances), but disagreement in the implied means is significantly lower for first- than for higher-order expectations (2.54 vs. 2.64; p = 0.04, Levene's test of homogeneity of variances). Our data thus indicates that disagreement regarding first-order and higher-order expectations is rather comparable. If anything, disagreement regarding first-order expectations, we find that disagreement regarding higher-order beliefs would be smaller than disagreement regarding first-order tespectations is greater than in higher-order inflation expectations. We cannot replicate this pattern in either sample or elicitation method.

⁶ To compare correlations, we use Cohen et al.'s (2014) procedure.

Result 3. CGKR Fact 3 not confirmed. The cross-sectional disagreement of first-order beliefs is lower than the disagreement of higher-order beliefs, both for households and firms.

3.1.3. Uncertainty

We calculate individual-level uncertainty for the probabilistic expectations measure. Our data shows that the average uncertainty is lower for first-order than for higher-order expectations, for both households and firms (households: 1.68 vs. 2.16, p < 0.01; firms: 1.22 vs .136, p < 0.01; Wilcoxon signed-rank tests).⁷ Another way to measure the degree of uncertainty inherent in expectations is to look at the correlation between point predictions and implied means from the probabilistic measure. For first-order expectations, we find that the correlation is 0.61 for households and 0.47 for firms. For higher-order expectations, the correlation is 0.47 for households and 0.39 for firms. Both for households and firms the correlation is stronger for first-order expectations (households: 0.61 vs. 0.47, p < 0.001; firms: 0.47 vs. 0.39, p < 0.001). That is, the coherence between different elicitation procedures is higher for first-order expectations than for higher-order expectations.

Our data stands in contrast to CGKR's, who find that average uncertainty in first-order inflation expectations is greater than in higher-order inflation expectations. In fact, our results point in the opposite direction in a consistent manner for both households and firms. Note, however, that in CGKR's follow-up wave (see their Table II), they find the same pattern as in our data, i.e., that uncertainty of higher-order inflation expectation is higher than the uncertainty of first-order inflation expectations.

Result 4. CGKR Fact 4 is not confirmed. Both for households and firms, the average uncertainty on the individual level is lower for first- than for higher-order expectations.

⁷ Note that the relation between uncertainty of first-order and higher-order beliefs determines the shape of the red line in Panel B of Figure 2. Specifically, in our data, the average of the individual-level distributions of first-order beliefs is less dispersed than the corresponding distribution of higher-order beliefs. Figure D2 in Appendix D depicts the same data as Figure 2, Panel B, but for different household and firm subgroups.

3.1.4. Comparison between Disagreement and Uncertainty

We find that the average degree of uncertainty is consistently lower than the cross-sectional dispersion or variance. This is true for both disagreement in point predictions and disagreement in implied means, and holds for first- and higher-order expectations and for both households and firms (households: first-order: point prediction 2.08 / implied mean 2.78 vs. uncertainty 1.68; higher-order: point prediction 2.21 / implied mean 2.78 vs. uncertainty 2.16; firms: first-order: point prediction 1.57 / implied mean 2.54 vs. uncertainty 1.22; higher-order: point prediction 1.55 / implied mean 2.64 vs. uncertainty 1.36). All differences are statistically significant at the 0.1%-level (Wilcoxon signed-rank tests). This finding is consistent with the results reported by CGKR.

Result 5. CGKR Fact 5 is confirmed. The average level of uncertainty is smaller than the crosssectional dispersion in inflation expectations for both households and firms, for both point and probabilistic expectations and for both first- and higher-order expectations.

Two key findings emerge from our analyses in Section 3.1. First, we only replicate facts 1, 2, and 5 of CGKR, and find opposite patterns for fact 3 and 4. Second, our pattern emerges highly consistently for our large samples of firms and households. We discuss the implications in Section 4. In Section 3.2. and 3.3. we take a closer look at the determinants of the prediction errors in higher-order expectations, and the role of heterogeneity in firm and household characteristics.

3.2 Prediction Error of Higher-Order Inflation Expectations

Next, we are interested in how accurate households and firms are in predicting the mean expectations of others. We analyze the individual prediction error, measured as the difference between individual higher-order beliefs and the actual mean first-order expectation. The smaller the difference, the more accurate the prediction. Depending on the elicitation method, we either use the average of the point predictions or the average of the implied means to determine the actual mean first-order expectation. Figure 3 shows the distribution of the prediction error on the population level. To compare the magnitude of prediction error, we use the absolute value, thereby neglecting the direction of error (i.e., whether higher-order beliefs overestimate or underestimate actual average first-order expectations). We find that predictions are better for firms than for households. Households exhibit an absolute value of prediction error of 1.44 (1.85) in the point (probabilistic) measure. Firms exhibit an absolute value of prediction error of 1.08 (1.71) in the

point (probabilistic) measure. Firms are less prone to error for both elicitation methods (point: 1.44 vs. 1.08, p < 0.001; probabilistic: 1.84 vs. 1.71, p < 0.001, Mann-Whitney-U-Tests). Moreover, comparing the two elicitation methods shows that both for households and firms, higher-order point expectations are less prone to error than higher-order probabilistic expectations (households: 1.44 vs. 1.85, p < 0.001; probabilistic: 1.08 vs. 1.71, p < 0.001, Wilcoxon signed-rank tests). However, one needs to keep in mind Result 1, which shows that on average neither households nor firms overestimate or underestimate the actual mean first-order inflation expectations.

Result 6. Higher-order expectations are more accurate for firms than for households, and more accurate for point predictions than for implied means derived from the probabilistic measure.



Figure 3. Prediction error of higher-order expectations

We are also interested in the relation between individual level uncertainty and prediction error (Table 2). Accordingly, we conduct correlation analysis between uncertainty and the absolute value of the prediction error. The positive correlations show that the more uncertain households and firms are, the greater is the prediction error of their higher-order expectations. Moreover, the relation

Notes: Prediction error is measured as the difference between the higher-order belief of an individual respondent and the actual mean first-order expectation (i.e., higher-order belief minus mean first-order belief).

between uncertainty in (both first-order and higher-order) inflation expectations and the extent of prediction error is stronger in households than in firms.

Result 7. Uncertainty and prediction error are positively correlated. The relation between uncertainty and the prediction error is stronger in households than in firms.

	House	holds	Firms		
	Uncertainty of first- order expectation	Uncertainty of higher-order expectation	Uncertainty of first-order expectation	Uncertainty of higher- order expectation	
Prediction error of point inflation expectation	0.22***	0.23***	0.07***	0.05**	
Prediction error of probabilistic inflation expectation	0.17***	0.10***	0.05**	0.03	

Table 2. Correlation between uncertainty and prediction error

Notes: In this analysis, prediction error is measured as the absolute difference between the higher-order belief of an individual and the actual mean first-order expectation. The smaller that difference, the better the prediction. We report Pearson's correlation coefficients. *, **, *** indicates significance at the 5%, 1%, and 0.1% level.

3.3 Sample Splits and Household and Firm Characteristics

We next make use of the available household and firm characteristics to (i) probe whether results 1-5 are robust to sample splits, and to (ii) study whether characteristics affect expectations and uncertainty about expectations. We find that the reported results regarding the relation between first-order and higher-order expectations consistently occur across various household and firm subsamples (see Table A1 for households and Table A2 for firms in Appendix A). While the results thus replicate in each subsample, expectations may still be affected by heterogeneity in household and firm characteristics. Table 3 contains regression analyses of mean inflation expectations and uncertainty on household characteristics. The results indicate that means of both first-order and higher-order inflation expectations tend to be higher for female respondents. This is consistent with Bruine de Bruin et al. (2010) and D'Acunto et al. (2021b), who find that being female is associated with higher first-order inflation expectations. Moreover, consistent with Das et al. (2020), we find that first-order and higher-order inflation expectations decrease with income and are also lower if respondents have a university degree. There is no association with age. The results further show that uncertainty is positively related to being female, but negatively related to age and income.

University education does not seem to affect uncertainty. There are no qualitative differences between the results for first-order and higher-order expectations.

Table 4 contains regression analyses of mean inflation expectations and uncertainty on firm characteristics. Mean point expectations (both first-order and higher-order) are decreasing in the number of employees and are lower in West Germany than in East Germany. These differences, however, do not show up for the probabilistic measure. Firm characteristics also do not seem to be associated with differences in uncertainty.

Result 8. The relation between first-order and higher-order inflation expectations is robust to sample splits. Household characteristics affect first- and higher-order expectations in a consistent way. We identify no effects of firm characteristics on expectations.

	Mean point expectations		Mean pro expect	babilistic ations	Uncertainty		
	FO	НО	FO	HO	FO	НО	
Female	0.161	0.411***	0.353**	0.396**	0.204*	0.352***	
	(0.098)	(0.106)	(0.129)	(0.129)	(0.086)	(0.096)	
Age	0.001	0.004	0.003	0.009*	-0.009***	-0.017***	
	(0.003)	(0.003)	(0.004)	(0.004)	(0.003)	(0.003)	
Household income	-0.091***	-0.064***	-0.089***	-0.076**	-0.080***	-0.069***	
	(0.019)	(0.019)	(0.024)	(0.024)	(0.015)	(0.017)	
University degree	-0.346***	-0.233*	-0.302**	-0.363**	-0.140	0.020	
	(0.089)	(0.095)	(0.115)	(0.115)	(0.078)	(0.087)	
Constant	3.333***	2.977***	2.737***	2.266***	2.728***	3.490***	
	(0.246)	(0.246)	(0.332)	(0.323)	(0.205)	(0.229)	
Ν	2,224	2,224	2,233	2,233	2,233	2,233	

Table 3. Household	Characteristics and	Inflation Ex	pectations
--------------------	---------------------	--------------	------------

Notes: OLS regressions. *, **, *** indicates significance at the 5%, 1%, and 0.1% level. FO = first-order, HO = higher-order. Female is an indicator for female gender. Age is an integer (minimum is 16, maximum is 80). Household income has 13 categories (with category 1 referring to income between $0 \in$ and 500 \in and category 13 referring to income above 10,000 \in). College education is an indicator for a university degree. Robust standard errors are clustered on the individual level and reported in parentheses.

	Mean point expectations		Mean pro expect	babilistic ations	Uncertainty		
	FO	НО	FO	HO	FO	HO	
Employees	-0.192***	-0.189***	-0.118	-0.132	-0.001	0.016	
	(0.040)	(0.043)	(0.066)	(0.076)	(0.027)	(0.029)	
Turnover	0.032	0.062	0.035	0.032	0.015	-0.001	
	(0.038)	(0.041)	(0.061)	(0.070)	(0.027)	(0.029)	
Services	-0.076	-0.023	0.164	0.110	0.010	0.019	
	(0.064)	(0.064)	(0.106)	(0.108)	(0.045)	(0.048)	
West Germany	-0.313**	-0.266*	-0.027	-0.216	0.029	0.061	
	(0.101)	(0.104)	(0.179)	(0.180)	(0.070)	(0.072)	
Constant	4.744***	4.634***	3.590***	3.681***	1.154***	1.262***	
	(0.122)	(0.130)	(0.226)	(0.228)	(0.087)	(0.090)	
Ν	2,513	2,513	2,548	2,548	2,548	2,548	

Table 4. Firm Characteristics and Inflation Expectations

Notes: OLS regressions. *, **, *** indicates significance at the 5%, 1%, and 0.1% level. FO = first-order, HO = higher-order. Employees is the number of employees and consists of five categories with category 1 referring to 1-10 employees and category 5 referring to more than 1,000 employees. Turnover consists of five categories with category 1 referring to an annual turnover below 1 million Euro and category 5 referring to an annual turnover above 229 million Euro. Services is an indicator for the firm belonging to the services sector (rather than the industrial or the construction sector). West Germany is an indicator for the firm being based in West Germany (rather than East Germany). Robust standard errors are clustered on the individual level and reported in parentheses.

Tables 5 and 6 show regression analyses of the absolute value of prediction error on household and firm characteristics, respectively. For households, the prediction error is positively associated with being female, but negatively with age, household income and having a university degree. For firms, the prediction error decreases with the number of employees, and is smaller for firms located in West Germany.

Table 5. Hous Pre	ehold Charac ediction Error	teristics and	Table 6. Firm Characteristics and Prediction Error			
	Point	Probabilistic		Point	Probabilistic	
Female	0.352*** (0.081)	0.427*** (0.096)	Employees	-0.0756* (0.0303)	-0.140* (0.0583)	
Age	-0.008*** (0.002)	-0.002 (0.003)	Turnover	0.0176 (0.0291)	0.0189 (0.0515)	
Household income	-0.030* (0.014)	-0.047** (0.018)	Services	-0.0517 (0.0458)	-0.169* (0.0817)	
University degree	-0.153* (0.072)	-0.386*** (0.084)	West Germany	-0.244** (0.0744)	-0.420** (0.134)	
Constant	2.020*** (0.182)	2.309*** (0.244)	Constant	1.444*** (0.0937)	2.420*** (0.171)	
Ν	2,224	2,233	Ν	2,513	2,548	
Notes: OLS regressio	ons. *, **, *** in	dicates	Notes: OLS regres	sions. *, **, ***	indicates	

Table 5. Household Characteristics	and
Prediction Error	

significance at the 5%, 1%, and 0.1% level. FO = firstorder, HO = higher-order. Female is an indicator for female gender. Age is an integer (minimum is 16, maximum is 80). Household income has 13 categories (with category 1 referring to income between 0€ and 500€ and category 13 referring to income above 10,000€). College education is an indicator for a university degree. Robust standard errors are clustered on the individual level and reported in parentheses.

significance at the 5%, 1%, and 0.1% level. FO = first-order, HO = higher-order. Employees is the number of employees and consists of five categories with category 1 referring to 1-10 employees and category 5 referring to more than 1,000 employees. Turnover consists of five categories with category 1 referring to an annual turnover below 1 million Euro and category 5 referring to an annual turnover above 229 million Euro. Services is an indicator for the firm belonging to the services sector (rather than the industrial or the construction sector). West Germany is an indicator for the firm being based in West Germany (rather than East Germany). Robust standard errors are clustered on the individual level and reported in parentheses.

4. Implications for calibrating noisy information models

The previous section reveals several relationships between first- and higher-order inflation expectations that are remarkably consistent between firms and households. At the same time, we find that some of the documented relationships are inconsistent with the stylized facts of CGKR.

Both disagreement and uncertainty in higher-order expectations are greater than in first-order expectations, which is the opposite of what CGKR find.

In this section, we investigate this discrepancy through an extension of the static noisy information model of Morris and Shin (2002). To reconcile their stylized facts about higher-order inflation expectations with theoretical predictions in settings where firms perform infinite regress in their expectations, CGKR consider multiple extensions of the baseline noisy information model. A promising extension is a model with heterogeneity in long-run priors (Patton and Timmermann, 2010), which we consider in this section. The idea is that forecasters shrink their optimal inflation forecasts toward their prior beliefs about long-run inflation which can reconcile most of CGKR's facts. We focus only on its predictions regarding the relative magnitude of disagreement and uncertainty in first- and higher-order expectations. We present the key intuitions here and refer for detailed derivations to Appendix E.

Suppose firm $i \in [0,1]$ chooses to set its optimal price, p_i , as a linear combination of its expectation of a fundamental, m, and its expectation of the aggregate price level in the economy, \bar{p} :

(1)
$$p_i = (1 - \alpha)E_i[m] + \alpha E_i^*[\bar{p}],$$

(2)

where $\alpha \in (0,1)$ describes the degree of complementarity in pricing. Additionally, assume that $\bar{p} \stackrel{\text{def}}{=} \int_0^1 p_j dj$, such that individual *i* can iterate the optimal price equation forward by substituting the average optimal price equation for the aggregate price level:

$$p_i = (1 - \alpha)E_i[m] + \alpha E_i^* \left[\int_0^1 p_j dj\right],$$

Following Patton and Timmermann (2010), we allow individual's "long-run" prior, μ_i , to skew expectations of the aggregate price level in the economy:

(3)
$$E_{i}^{*}[\bar{p}] = \omega \mu_{i} + (1 - \omega)E_{i}[\bar{p}]$$
$$= \omega \mu_{i} + (1 - \omega)\{(1 - \alpha)E_{i}[\bar{E}[m]] + \alpha E_{i}[\bar{E}[\bar{p}]]\},$$

where $E_i^*[\bar{p}]$ denotes the skewed first-order expectation of \bar{p} and $E_i[\bar{p}] = (1 - \alpha)E_i[\bar{E}[m]] + \alpha E_i[\bar{E}[\bar{p}]]$ denotes the rational (non-skewed) expectation of \bar{p} . Additionally, $\mu_i \sim N(\bar{\mu}, \kappa_{\mu}^{-1})$, with $\bar{\mu}$ and κ_{μ}^{-1} measuring the average level and the dispersion of the "long-run" priors, and $\omega = \frac{Var(E_i[\int p_j d_j])}{\gamma^2 + Var(E_i[\int p_j d_j])}$ with $\gamma^2 \ge 0$ being a parameter measuring the degree to which an individual prefers her own "long-run" prior. Further, define the average expectation in the economy for variable m as $\bar{E}[m]$ and let $E_i[\bar{E}[m]]$ be the expectation of individual i about the average

expectation in the economy. Similarly, let $E_i(\bar{p})$ denote the first-order expectation about the average price level, and $E_i[\bar{E}[\bar{p}]]$ the second-order expectation about the price level. We can iterate these expectations to the *k*th higher-order recursively: $\bar{E}^k[X] = \int_0^1 E_j(\bar{E}^{k-1}[X]) dj$.

To characterize how individuals form (higher-order) expectations about the fundamental, we follow CGKR and assume that individuals do not possess full information. Instead of observing *m* perfectly, they receive one noisy public signal and one private signal. Each signal individually reflects the true value of *m* combined with some noise. The public signal $y = m + \varepsilon$, where $\varepsilon \sim N(0, \kappa_y^{-1})$ is common across firms and the private signal takes the form: $x_i = m + v_i$, with $v_i \sim N(0, \kappa_x^{-1})$, and where κ_x and κ_y denote the precision of each type of signal. In order to obtain an individual expectation of *m*, firms weight their signals according to the relative noise in each:

(4)
$$E_i[m] = \frac{\kappa_y}{\kappa_x + \kappa_y} y + \frac{\kappa_x}{\kappa_x + \kappa_y} x_i = (1 - \delta)y + \delta x_i$$

where $\delta = \frac{\kappa_x}{\kappa_x + \kappa_y}$. The intuition is straightforward: As the private signal becomes more precise relative to the public signal, the firm places relatively more weight on it in when forming beliefs about the fundamental. It can be shown that firm *i*'s higher-order expectation about the average expectation of other firms is (with $\overline{E}^k[X]$ denoting the *k*th higher-order belief):

(5)
$$E_i[\bar{E}^k[m]] = (1 - \delta^{k-1})y + \delta^{k-1}E_i[\bar{E}^{k-1}[m]] = (1 - \delta^k)y + \delta^k x_i$$

Equation (5) shows that higher order expectations will depend increasingly more on the public signal as that signal is common across firms.

By repeatedly substituting into the optimal pricing equation (1), one can show that the aggregate price level also becomes an average of increasingly higher-order expectations of the fundamental:

(6)
$$p_{i} = \alpha \omega \mu_{i} + \alpha \omega \sum_{k=0}^{\infty} \alpha^{k+1} (1-\omega)^{k+1} E_{i} \left[\overline{E}^{k} [\overline{\mu}] \right] + (1)$$
$$-\alpha) \sum_{k=0}^{\infty} \alpha^{k} (1-\omega)^{k} \left[(1-\delta^{k+1})y + \delta^{k+1} x_{i} \right]$$

An individual firm sets its price as a function of its "long-run" prior μ_i , a sum of progressively higher-order expectations of the average prior $\bar{\mu}$, and a sum of progressively higher-order expectations of the fundamental *m*. Following CGKR, we assume that individuals do not know the true average prior $\bar{\mu}$, but rather observe a private signal of the mean: $\zeta_i \sim N(\bar{\mu}, \kappa_{\zeta}^{-1})$. The uncertainty about the aggregate prior, κ_{ζ}^{-1} , is necessary to bring priors and the dispersion they provide into higher-order expectations. Additionally, with heterogeneity in priors, there are two sources of variation between first- and higher-order expectations. On the one hand, information in private signals differs across individuals. On the other hand, there is a potential discrepancy in individual's own prior beliefs (regarding the long-run prior) and the prior belief they assign to others. Because both sources of uncertainty are assumed to be uncorrelated and the weights on signals are different for first- and higher-order expectations, the cross-sectional correlation between first- and second-order beliefs is imperfect, which is consistent with our stylized fact 1.

Finally, we can proceed to find expressions for cross-sectional disagreement and forecast uncertainty in first- and second-order expectations (see Appendix E). It can be shown that disagreement in second-order expectations ($Var\left[E_i[\bar{p}]\right]$) is greater than disagreement in first-order expectations ($Var\left[E_i[\bar{p}]\right]$) if:

(7)

$$Var\left[E_{i}\left[\bar{E}\left[\bar{p}\right]\right]\right] > Var\left[E_{i}\left[\bar{p}\right]\right]$$

$$\Leftrightarrow \left(2(1-\omega)(\omega\theta\omega')(\omega'-1)+\omega^{2}(\omega'^{2}-2))\kappa_{\mu}^{-1}\right)$$

$$+ \left[\omega\left((\omega+(1-\omega)\theta)+((1-\omega)\theta)\right)\right](1)$$

$$- \omega')^{2}\kappa_{\zeta}^{-1}+(1-\omega)^{2}(\phi_{x}\delta)^{2}(\delta^{2}-1)\kappa_{x}^{-1}>0$$

Similarly, one can show that uncertainty in second-order expectations $(\Omega_{\{E_i[\bar{E}[\bar{p}]]|y\}})$ is greater than uncertainty in first-order expectations $(\Omega_{\{E_i[\bar{p}]|y\}})$ if:

(8)

$$\Omega_{\left\{E_{i}\left[\bar{E}\left[\bar{p}\right]\right]|y\right\}} > \Omega_{\left\{E_{i}\left[\bar{p}\right]|y\right\}}$$

$$\Leftrightarrow \left[\omega\left((\omega + (1-\omega)\theta) + ((1-\omega)\theta)\right)\right](1-\omega')^{2}\kappa_{\zeta}^{-1}$$

$$+ (1-\omega)^{2}(\phi_{x}\delta)^{2}(\delta^{2}-1)\kappa_{x}^{-1} > 0$$

In contrast to the basic noisy-information model with infinite regress which CGKR introduce, the extension with heterogenous long-run priors does not make a clear prediction on the relative magnitude of disagreement and uncertainty in first versus second-order expectations. Instead, the prediction depends on the relative magnitudes of κ_{ζ}^{-1} and κ_{x}^{-1} . In particular, if κ_{x}^{-1} and κ_{μ}^{-1} are large relative to κ_{ζ}^{-1} the model produces $Var\left[E_{i}[\bar{E}[\bar{p}]]\right] < Var[E_{i}[\bar{p}]]$, as well as $\Omega_{\{E_{i}[\bar{E}[\bar{p}]]|y\}} < \Omega_{\{E_{i}[\bar{p}]|y\}}$, which is the finding of CGKR. However, if κ_{x}^{-1} and κ_{μ}^{-1} are small relative to κ_{ζ}^{-1} , the model produces $Var\left[E_{i}[\bar{p}]\right]$ as well as $\Omega_{\{E_{i}[\bar{p}]|y\}} > \Omega_{\{E_{i}[\bar{p}]|y\}}$, which is

consistent with our findings. To interpret the directions of both predictions, it is instructive to review what the parameters capture. The parameter κ_x^{-1} measures dispersion in private signals. For instance, when κ_x^{-1} is high, there is a lot of dispersion in private signals. Moving from first-order to second-order beliefs, individuals place less weight on their own priors (and on the private signal), which tends to reduce disagreement and uncertainty in second-order relative to first-order beliefs. κ_{μ}^{-1} captures dispersion in individuals' long-run priors. A high κ_{μ}^{-1} indicates a lot heterogeneity in long-run priors about average prices.

Consider an individual who understands that her first-order belief is skewed by her long-run prior and that others also report skewed first-order beliefs. When thinking about other individuals' beliefs, each individual tries to remove her own "bias" (i.e., her own long-run prior) from the reported value. This tends to reduce dispersion in second-order beliefs relative to first-order beliefs. Finally, κ_{ζ}^{-1} measures dispersion in beliefs about other individuals' long-run priors (i.e. uncertainty about the aggregate prior). When κ_{ζ}^{-1} is high, there is a lot of dispersion in beliefs about other individuals' long-run priors. This tends to increase both uncertainty and disagreement in second-order forecasts relative to first-order forecasts, which is opposite to the effect of κ_{μ}^{-1} . Reproducing our facts would require that the effect of κ_{x}^{-1} and κ_{μ}^{-1} be weaker than the effect of κ_{ζ}^{-1} . Conversely, if the effect of κ_{x}^{-1} and κ_{μ}^{-1} is stronger, than the model would reproduce the facts of CGKR.

It is important to note that the relation between uncertainty and disagreement in first- versus higher-order expectations always moves in tandem as they rely on the same conditions: If one finds that disagreement in second-order expectations is higher than disagreement in first-order expectations, one should also find that uncertainty is higher in second-order than in first order expectations. Our finding of higher uncertainty and disagreement in second-order than in first-order expectations is thus consistent with these theoretical predictions but implies different beliefs about long-run priors than the results of CGKR. This reconciles the differences in results between our study and the study by CGKR. Importantly, to generate predictions that match our data, dispersion in beliefs about others' long-run priors needs to be high relative to dispersion in private signals and to dispersion in individuals' own long-run priors. Intuitively, this seems plausible. It suggests that uncertainty aggregates when individuals extrapolate from they believe to what others believe.

To gain some insight into potential long-run priors, we consider a question of our household panel (BOP-HH) on long-run inflation expectations. For Wave 10 and from Wave 13 onwards, BOP-HH includes questions on the expected inflation rate in 10 years. The question is framed as a point prediction. Given the long time frame, it is unlikely that people have reliable signals to base their expectations on and we argue that these long-run expectations capture long-run priors for inflation expectations. Figure 5 Panel A shows the mean of 10-year inflation expectations, in comparison to the 12 months point prediction of the first-order inflation expectation. Panel B displays disagreement.

Figure 5. Comparison of long-run priors and 12 months point predictions



Panel B. Disagreement (BOP-HH)



We make two observations. First, 10-year expectations are much more stable than the 12 months expectations. While the 12 month point prediction shows a clear upwards trend over the depicted time frame, long term predictions are mostly flat. Second, results in Panel B display that long-run priors show considerable disagreement and indeed somewhat more disagreement than the point

prediction for households. This further supports the existence of heterogenous long-term priors as suggested by CGKR and Patton and Timmermann, (2010). The introduction of heterogenous long-run priors appears to be a promising candidate to rationalize differences between our results and those of CGKR. Future work could fruitfully examine expectations about the long-run priors of others, and how uncertainty and disagreement in these second-order expectations compare to their own long-run priors. This would identify $E_i[\bar{\mu}]$, and in particular κ_c^{-1} .

5. Conclusion

This article presents novel survey evidence on firms' and households' higher-order inflation expectations. Despite playing an important role in macroeconomic and intertemporal microeconomic models, empirical evidence on them is scarce. Our study provides the first evidence on how households form higher-order beliefs about inflation and how such beliefs compare to their first-order expectations. Additionally, having access to both household and firm first- and higher-order expectations data, we can identify common patterns that allow us to establish a number of stylized facts. These insights can be used to discipline models of higher-order beliefs. For instance, a central implication of both our samples is that uncertainty accrues as individuals extrapolate from their beliefs to what others might know. This is in contrast to the assumption of the baseline noisy information model, which assumes that uncertainty is lower in higher order-expectations as agents put higher weight on a public signal when forming their beliefs. One potential avenue to rationalize this discrepancy is by introducing heterogenous long-run priors which add another source of uncertainty that is not present in first-order expectations. To test such a conjecture empirically, future research would not only need to assess agents' higher-order expectations, but also their uncertainty regarding the long-run priors of others.

Beyond implications for theory, our results also offer more immediate policy guidance. For instance, forward guidance typically aims not only at moving first-order but also higher order-expectations. The common patterns between first- and higher-order expectations among firms and households we identified suggest that such communication-based policy tools are not only useful for moving expectations of professional forecasters or firm managers but also for the average household.

References

- Angeletos, G. M., &, La'O, J. (2009). Incomplete information, higher-order beliefs and price inertia. *Journal of Monetary Economics*, 56, S19-S37.
- Angeletos, G. M., & Lian, C. (2018). Forward guidance without common knowledge. *American Economic Review*, 108(9), 2477-2512.
- Angeletos, G. M., Huo, Z., & Sastry, K. A. (2021). Imperfect macroeconomic expectations: Evidence and theory. *NBER Macroeconomics Annual*, 35(1), 1-86.
- Bacchetta, P., & Van Wincoop, E. (2008). Higher order expectations in asset pricing. *Journal of Money, Credit and Banking*, 40(5), 837-866.
- Bruine de Bruin, W., Vanderklaauw, W., Downs, J. S., Fischhoff, B., Topa, G., & Armantier, O. (2010). Expectations of inflation: The role of demographic variables, expectation formation, and financial literacy. *Journal of Consumer Affairs*, *44*(2), 381-402.
- Cohen, P., West, S. G., & Aiken, L. S. (2014). Applied multiple regression/correlation analysis for the behavioral sciences. Psychology press.
- Coibion, O., Gorodnichenko, Y., & Kumar, S. (2018). How do firms form their expectations? New survey evidence. *American Economic Review*, *108*(9), 2671-2713.
- Coibion, O., Gorodnichenko, Y., & Ropele, T. (2020a). Inflation expectations and firm decisions: New causal evidence. *The Quarterly Journal of Economics*, 135(1), 165-219.
- Coibion, O., Gorodnichenko, Y., Kumar, S., & Pedemonte, M. (2020b). Inflation expectations as a policy tool?. *Journal of International Economics*, 124, 103297.
- Coibion, O., Gorodnichenko, Y., Kumar, S., & Ryngaert, J. (2021). Do You Know that I Know that You Know...? Higher-Order Beliefs in Survey Data. *The Quarterly Journal of Economics*, 136(3), 1387-1446.
- D'Acunto, F., Malmendier, U., Ospina, J., & Weber, M. (2021a). Exposure to grocery prices and inflation expectations. *Journal of Political Economy*, 129(5), 1615-1639.
- D'Acunto, F., Malmendier, U., & Weber, M. (2021b). Gender roles produce divergent economic expectations. *Proceedings of the National Academy of Sciences*, *118*(21), e2008534118.
- Das, S., Kuhnen, C. M., & Nagel, S. (2020). Socioeconomic status and macroeconomic expectations. *The Review of Financial Studies*, 33(1), 395-432.
- Farhi, E., & Werning, I. (2019). Monetary policy, bounded rationality, and incomplete markets. *American Economic Review*, 109(11), 3887-3928.
- Gabaix, X. (2020). A behavioral New Keynesian model. *American Economic Review*, 110(8), 2271-2327.

- Hellwig, C., & Venkateswaran, V. (2009). Setting the right prices for the wrong reasons. *Journal* of Monetary Economics, 56, S57-S77.
- Mankiw, N. G., & Reis, R. (2002). Sticky information versus sticky prices: a proposal to replace the New Keynesian Phillips curve. *The Quarterly Journal of Economics*, 117(4), 1295-1328.
- Mankiw, N. G., Reis, R., & Wolfers, J. (2003). Disagreement about inflation expectations. *NBER Macroeconomics Annual*, 18, 209-248.
- Manski, C. F. (2004). Measuring expectations. *Econometrica*, 72(5), 1329-1376.
- Morris, S., & Shin, H. S. (1998). Unique equilibrium in a model of self-fulfilling currency attacks. *American Economic Review*, 587-597.
- Morris, S., & Shin, H. S. (2002). Social value of public information. *American Economic Review*, 92(5), 1521-1534.
- Nimark, K. (2008). Dynamic pricing and imperfect common knowledge. *Journal of monetary Economics*, 55(2), 365-382.
- Patton, A. J., & Timmermann, A. (2010). Why do forecasters disagree? Lessons from the term structure of cross-sectional dispersion. *Journal of Monetary Economics*, 57(7), 803-820.
- Rich, R., & Tracy, J. (2010). The relationships among expected inflation, disagreement, and uncertainty: evidence from matched point and density forecasts. *The Review of Economics and Statistics*, 92(1), 200-207.
- Vellekoop, N., & Wiederholt, M. (2019). Inflation Expectations and Choices of Households. SAFE Working Paper No. 250.
- Woodford, M. (2002). Inflation stabilization and welfare. *Contributions to Macroeconomics*, 2(1), 1009.

Appendix A: Additional Tables

Panel A. Poin	t expectations by househo	olds							
		Obs.	M	ean	Disagr	eement	Uncer	tainty	Correlation
		(1)	(2	2)	(.	3)	(4	4)	(5)
			FO	HO	FO	HO	FO	HO	
All		2,306	2.65	2.75	2.08	2.21	-	-	0.59
Gender	Male	1,417	2.53	2.55	1.79	1.92	-	-	0.60
	Female	889	2.85	3.07	2.46	2.57	-	-	0.64
Age	18-39	361	2.53	2.60	2.09	2.29	-	-	0.61
	40-59	870	2.68	2.78	2.10	2.34	-	-	0.56
	>59	1,069	2.65	2.77	2.02	2.06	-	-	0.61
Income	<2499	579	3.10	3.17	2.44	2.36	-	-	0.57
	2500-4999	1,172	2.60	2.66	2.01	2.18	-	-	0.57
	>4999	483	2.28	2.50	1.57	1.94	-	-	0.63
Education	No college degree	1,258	2.88	2.92	2.23	2.34	-	-	0.61
	College degree	1,034	2.36	2.53	1.79	2.01	-	-	0.56
Panel B. Prob	abilistic expectations by h	nouseholds							
		Obs.	M	ean	Disagr	eement	Uncer	tainty	Correlation
		(1)	(2	2)	(.	3)	(4	4)	(5)
			FO	HO	FO	HO	FO	HO	
All		2,310	2.24	2.20	2.78	2.78	1.68	2.16	0.67
Gender	Male	1,401	2.07	2.02	2.31	2.38	1.57	2.01	0.58
	Female	909	2.49	2.47	3.36	3.28	1.85	2.41	0.69
Age	18-39	371	1.89	1.73	2.98	2.56	1.89	2.58	0.57
-	40-59	885	2.40	2.28	2.92	2.88	1.73	2.32	0.67
	>59	1,048	2.21	2.29	2.56	2.74	1.55	1.88	0.72
Income	<2499	583	2.54	2.55	3.29	3.23	1.92	2.36	0.71
	2500-4999	1,190	2.26	2.19	2.73	2.69	1.67	2.13	0.64
	>4999	470	1.84	1.82	1.86	2.15	1.38	1.96	0.60
Education	No college degree	1,283	2.46	2.45	3.02	3.10	1.79	2.21	0.67
	College degree	1 014	196	1 89	2 36	2 21	1 53	2.09	0.66

Table A1. Household's moments of first- and higher-order inflation expectations by subsamples

Notes: FO = first-order, HO = higher-order. The table reports basic moments of first-order and higher-order inflation expectations. In order to avoid extreme values, the data on point expectations in Panel 3A is truncated below -12 and above +12. The implied mean from the probabilistic estimation in Panel 3B, column (2) is calculated using a value of -16 and 16, respectively, for the outer bins. Disagreement in column (3) reports the cross-sectional standard deviation of mean inflation forecasts. Uncertainty in column (4) refers to the standard deviation of the reported probability distribution for future inflation using a value of -16 and 16, respectively, for the outer bins. Column (5) reports the correlation between first-order and higher-order expectations.

Panel A. Point	t expectations by firms								
		Obs.	M	ean	Disagı	reement	Uncer	tainty	Correlation
		(1)	(2	2)	(3)	(4	4)	(5)
			FO	HO	FO	HO	FO	HO	
All		2,681	4.09	4.13	1.57	1.55	-	-	0.71
Employees	1-10	963	4.23	4.23	1.66	1.58	-	-	0.71
	11-50	861	4.20	4.21	1.67	1.67	-	-	0.69
	51-200	498	3.91	3.99	1.35	1.41	-	-	0.72
	201-1000	252	3.72	3.82	1.26	1.31	-	-	0.76
	>1000	79	3.63	3.70	1.15	0.98	-	-	0.75
Turnover	<1 mio	864	4.14	4.14	1.59	1.54	-	-	0.69
	1mio-7 mio	775	4.19	4.22	1.68	1.71	-	-	0.70
	7 mio-34 mio	538	4.08	4.11	1.50	1.39	-	-	0.72
	34 mio-229 mio	323	3.84	3.92	1.34	1.41	-	-	0.74
	>229 mio	134	3.83	4.02	1.32	1.51	-	-	0.82
Sector	Industry	720	4.08	4.10	1.59	1.57	-	-	0.76
	Construction	287	4.17	4.16	1.51	1.60	-	-	0.66
	Services	1,525	4.08	4.13	1.55	1.55	-	-	0.71
Region	West Germany	2,254	4.06	4.09	1.53	1.51	-	-	0.71
-	East Germany	331	4.37	4.35	1.74	1.81	-	-	0.78
Panel B. Proba	abilistic expectations by	firms							
		Obs.	M	ean	Disagi	reement	Uncer	tainty	Correlation
		(1)	(2	2)	(3)	(4	4)	(5)
			FO	HO	FO	HO	FO	HO	
All		2,718	3.51	3.37	2.54	2.64	1.22	1.36	0.68
Employees	1-10	976	3.54	3.44	2.85	2.83	1.20	1.33	0.67
1 2	11-50	883	3.68	3.46	2.52	2.66	1.24	1.39	0.73
	51-200	500	3.40	3.36	2.13	2.29	1.21	1.34	0.74
	201-1000	253	3.06	2.96	2.26	2.46	1.28	1.40	0.61
	>1000	80	3.33	2.86	1.53	2.34	1.29	1.49	0.18
Turnover	<1 mio	877	3.47	3.36	2.78	2.74	1.21	1.34	0.71
	1mio-7 mio	795	3.64	3.46	2.62	2.64	1.22	1.38	0.76
	7 mio-34 mio	544	3.52	3.31	2.32	2.62	1.22	1.35	0.68
	34 mio-229 mio	321	3.25	3.13	2.16	2.38	1.22	1.36	0.54
	>229 mio	135	3.42	3.36	1.78	2.28	1.34	1.44	0.69
Sector	Industry	729	3.29	3.15	2.87	2.84	1.25	1.37	0.72
	Construction	295	3.62	3.57	2.35	2.53	1.14	1.28	0.77
	Services	1,545	3.57	3.41	2.43	2.57	1.22	1.36	0.64
					the set of				
Region	West Germany	2,280	3.49	3.32	2.45	2.57	1.23	1.37	0.65

Table A2. Firm's moments of first- and higher-order inflation expectations by subsamples

Notes: FO = first-order, HO = higher-order. The table reports basic moments of first-order and higher-order inflation expectations. In order to avoid extreme values, the data on point expectations in Panel 4A is truncated below -12 and above +12. The implied mean from the probabilistic estimation in Panel 4B, column (2) is calculated using a value of -16 and 16, respectively, for the outer bins. Disagreement in column (3) reports the cross-sectional standard deviation of mean inflation forecasts. Uncertainty in column (4) refers to the standard deviation of the reported probability distribution for future inflation using a value of -16 and 16, respectively, for the outer bins. Column (5) reports the correlation between first-order and higher-order expectations.

Appendix B: Survey Questions

Iten	1	Wording of Question and Input
1	One-year ahead first-order point	Do you think inflation or deflation is more likely over the next
	inflation expectations qualitative	twelve months?
		• Inflation more likely
		 Deflation more likely
1A	One-year ahead first-order point	What do you think the rate of inflation in Germany will roughly
	inflation expectations quantitative (if	be over the next twelve months?
	respondent states in the qualitative	
	question "Inflation more likely")	[Decimal number with one decimal place]
1B	One-year ahead first-order point	What do you think the rate of deflation in Germany will roughly
	deflation expectations quantitative (if	be over the next twelve months?
	respondent states in the qualitative	
	question "Deflation more likely")	[Decimal number with one decimal place]
2	One-year ahead first-order probabilistic	In your opinion, how likely is it that the rate of inflation will
	inflation expectations	change as follows over the next twelve months?
		[Subjects have to assign probability weights to the different bins
		that need to add up to 100%]
		a) The rate of deflation (opposite of inflation) will be 12% or higher.
		 b) The rate of deflation (opposite of inflation) will be between 8% and less than 12%.
		c) The rate of deflation (opposite of inflation) will be between 4% and less than 8%.
		 d) The rate of deflation (opposite of inflation) will be between 2% and less than 4%.
		 e) The rate of deflation (opposite of inflation) will be between 0% and less than 2%
		f) The rate of inflation will be between 0% and less than 2%
		g) The rate of inflation will be between 2% and less than 4%
		h) The rate of inflation will be between 4% and less than 8%
		i) The rate of inflation will be between 8% and less than 12%.
		i) The rate of inflation will be 12% or higher.
3	One-year ahead higher-order point	In your opinion, do the other participants in this survey believe
-	inflation expectations qualitative	that inflation or deflation is more likely in Germany over the
	1 1	next twelve months?
		• Inflation more likely
		• Deflation more likely

Table B1. Survey Questions for Households

inflation expectations quantitative (if respondent states in the qualitative question "Inflation more likely") In your opinion, what do the other participants in this survey think the rate of inflation will roughly be over the next twelve months? 3B One-year ahead higher-order point In your opinion, what do the other participants in this survey think the rate of inflation will roughly be over the next twelve months?	ey elve
 SA One-year anead higher-order point inflation expectations quantitative (if respondent states in the qualitative question "Inflation more likely") B One-year ahead higher-order point [Decimal number with one decimal place] 3B One-year ahead higher-order point [In your opinion, what do the other participants in this survey think the rate of inflation will roughly be over the next twelve months? [Decimal number with one decimal place] 	ey elve
initiation expectations qualitative (in respondent states in the qualitative question "Inflation more likely") IDecimal number with one decimal place] 3B One-year ahead higher-order point In your opinion, what do the other participants in this survey.	
question "Inflation more likely") [Decimal number with one decimal place] 3B One-year ahead higher-order point In your opinion, what do the other participants in this survey.	
(Decimal number with one decimal place) 3B One-year ahead higher-order point In your opinion, what do the other participants in this survey	
3B One-year ahead higher-order point In your opinion, what do the other participants in this survey	
22 one jeur aneue ingher order point in jour opinion, what do the outer paraelpanto in this bar eg	ey
deflation expectations quantitative (if think the rate of deflation will roughly be over the next twelve	elve
respondent states in the qualitative months?	
question "Deflation more likely")	
[Decimal number with one decimal place]	
4 One-year ahead higher-order In your opinion, how likely do the other participants in this	5
probabilistic inflation expectations survey think it is that the rate of inflation will change as follow	llows
over the next twerve months?	
[Subjects have to assign probability weights to the different bi	t bins
that need to add up to 100%]	
a) The rate of deflation (opposite of inflation) will be 12% or	6 or
higher.	
b) The rate of deflation (opposite of inflation) will be betwee	ween
8% and less than 12%.	
c) The rate of deflation (opposite of inflation) will be betwee 4% and less than 8%.	ween
d) The rate of deflation (opposite of inflation) will be betwee 2% and less than 4%.	ween
e) The rate of deflation (opposite of inflation) will be betwee 0% and less than 2%	ween
f) The rate of inflation will be between 0% and less than 2%	2%.
g) The rate of inflation will be between 2% and less than 4%.	4%.
h) The rate of inflation will be between 4% and less than 8%	8%.
i) The rate of inflation will be between 8% and less than 12%	12%.
j) The rate of inflation will be 12% or higher.	
5 Five-years ahead first-order point And what value do you think the rate of inflation or deflation	on
inflation expectations quantitative will take on average over the next five years?	
[Decimal number with one decimal place]	
6 Ten-years ahead first-order point And what value do you think the rate of inflation or deflation	on
inflation expectations quantitative will take on average over the next ten years?	
[Decimal number with one decimal place]	

Iter	m	Wording of Question and Input
1	One-year ahead first-order point inflation expectations quantitative	What do you expect the rate of inflation to be over the next twelve months?
2	One-year ahead first-order probabilistic inflation expectations	[Decimal number with one decimal place] In your opinion, how likely is it that the rate of inflation will change as follows over the next twelve months?
		[Subjects have to assign probability weights to the different bins that need to add up to 100%]
		a) The rate of deflation (opposite of inflation) will be 12% or higher.b) The rate of deflation (opposite of inflation) will be between
		 8% and less than 12%. c) The rate of deflation (opposite of inflation) will be between 4% and less than 8%.
		d) The rate of deflation (opposite of inflation) will be between 2% and less than 4%.e) The rate of deflation (opposite of inflation) will be between
		 0% and less than 2%. f) The rate of inflation will be between 0% and less than 2%. g) The rate of inflation will be between 2% and less than 4%. h) The rate of inflation will be between 4% and less than 8%. i) The rate of inflation will be between 8% and less than 12%. j) The rate of inflation will be 12% or higher.
3	One-year ahead higher-order point inflation expectations quantitative	What rate of inflation do you think other enterprises in Germany are expecting on average over the next twelve months?
		[Decimal number with one decimal place]
4	One-year ahead higher-order probabilistic inflation expectations	In your opinion, how likely do other enterprises in Germany think it is that the rate of inflation will change as follows over the next twelve months? Input:
		[Subjects have to assign probability weights to the different bins that need to add up to 100%]
		 a) The rate of deflation (opposite of inflation) will be 12% or higher. b) The rate of deflation (opposite of inflation) will be between
		b) The rate of deflation (opposite of inflation) will be between 8% and less than 12%.c) The rate of deflation (opposite of inflation) will be between
		4% and less than 8%.d) The rate of deflation (opposite of inflation) will be between
		 2% and less than 4%. e) The rate of deflation (opposite of inflation) will be between 0% and less than 2%.
		 f) The rate of inflation will be between 0% and less than 2%. g) The rate of inflation will be between 2% and less than 4%. h) The rate of inflation will be between 4% and less than 8%. i) The rate of inflation will be between 8% and less than 12%. j) The rate of inflation will be 12% or higher.

Table B2. Survey Questions for Firms

Item		Wording of Question and Input
5	Three-years ahead first-order point inflation expectations quantitative	What do you expect the rate of inflation to be on average over the next three years?
		[Decimal number with one decimal place]
6	Five-years ahead first-order point inflation expectations quantitative	What do you expect the rate of inflation to be on average over the next five years?
		[Decimal number with one decimal place]

Appendix C: Differences between Point and Probabilistic Expectations

Figure 1 shows the values of mean first-order expectations (Panel A) and disagreement (Panel B) over time. First, average expectations elicited via point estimation are consistently higher than the average implied point estimations derived from the probabilistic measure. Second, disagreement is lower among the point estimations than among the implied estimations. Table 1 and Figure C1 zoom in on these aspects and compare the average distribution of mean first-order and higher-order inflation expectations on the individual level for the respective points in time. The differences in mean first-order expectations are statistically significantly different (households: 2.65 vs. 2.24, p < 0.001; firms: 4.09 vs. 3.51, p < 0.001, Mann-Whitney-U-Test). Likewise, the differences in disagreement of first-order expectations are statistically significantly different (households: 2.08 vs. 2.78, p < 0.001; firms: 1.57 vs. 2.64, p < 0.001, Levene's test of homogeneity of variances). Conducting the same analyses for higher-order expectations across elicitation methods yields the same results. The results indicate that, both for households and firms, averages of first-order and higher-order and higher-order within each elicitation method, but differ across elicitation methods.

Our results add to the mixed results from the literature. Consistent with our results, Coibion et al. (2018) find that the point estimation tends to produce higher mean expectations than the implied mean derived from the probabilistic distribution. In contrast to our results, Rich and Tracy (2010) explicitly study the coherence between point and probabilistic inflation expectations and do not find strong and systematic differences.



Figure C1. Overview about distribution of first-order and higher-order inflation expectations

Notes: The figure shows point and probabilistic mean first-order (FO) and higher-order (HO) inflation expectations of households and firms. Household data from Panel A stems from March 2021, firm data from Panel B stems from Q4 2021.

Appendix D: Additional Figures

Figure D1. Probability distributions of first-order and higher-order inflation expectations



Notes: The red circles are the average probability assigned to a particular bin across all respondents and their sum equals 100%. The shaded areas show the share of respondents reporting a given probability range in a specific inflation interval. The white area refers to the category zero.











Inflation interval

A4. Difference by education

Là



Inflation interval



Panel B. FirmsB1. Difference by number of employees

Notes: The lines indicate the difference (first-order minus higher-order expectation) in probability assigned to a specific bin.

B2. Difference by turnover

Appendix E: Derivation of a Noisy Information Model with Heterogenous Long-Run Priors

In the following, we use a modified version of the noisy information model of Morris and Shin (2002) to demonstrate how the established relationship between first- and higher-order expectations in our surveys compare with theoretical predictions. The extension follows earlier work by Patton and Timmermann (2010) and introduces heterogeneity in prior beliefs. The derivations presented in this section directly build on a similar exercise reported in Coibion et al. (2021)'s online appendix.

Suppose firm $i \in [0,1]$ chooses to set its optimal price, p_i , as a linear combination of its expectation of a fundamental, m, and its expectation of the aggregate price level in the economy, \bar{p} :

(1)
$$p_i = (1 - \alpha)E_i[m] + \alpha E_i^*[\bar{p}],$$

where $\alpha \in (0,1)$ describes the degree of complementarity in pricing. Additionally, assume that $\bar{p} \stackrel{\text{def}}{=} \int_0^1 p_j dj$, such that individual *i* can iterate the optimal price equation forward by substituting the average optimal price equation for the aggregate price level:

(1')
$$p_{i} = (1 - \alpha)E_{i}[m] + \alpha E_{i}^{*}\left[\int_{0}^{1} p_{j}dj\right],$$

Following Patton and Timmermann (2010), we allow manager's "long-run" prior, μ_i , to skew expectations of the aggregate price level in the economy:

(2)
$$E_i^*[\bar{p}] = \omega \mu_i + (1 - \omega) E_i[\bar{p}]$$
$$= \omega \mu_i + (1 - \omega) \{ (1 - \alpha) E_i[\bar{E}[m]] + \alpha E_i[\bar{E}[\bar{p}]] \},$$

where $E_i^*[\bar{p}]$ denotes the skewed first-order expectation of \bar{p} , $E_i[\bar{p}] = (1 - \alpha)E_i[\bar{E}[m]] + \alpha E_i[\bar{E}[\bar{p}]]$ denotes the mathematically correct non-skewed expectation of \bar{p} . Additionally, $\mu_i \sim N(\bar{\mu}, \kappa_{\mu}^{-1})$, and $\omega = \frac{Var(E_i[\int p_j d_j])}{\gamma^2 + Var(E_i[\int p_j d_j])}$ with $\gamma^2 \ge 0$ being a parameter measuring the degree to which a firm manager prefers her own "long-run" prior, and $\bar{\mu}$ and κ_{μ}^{-1} measuring the average level and the dispersion of the "long-run" priors. Further, define the average expectation in the economy for variable m as $\bar{E}[m]$ and let $E_i[\bar{E}[m]]$ be the expectation of individual i about the average expectation in the economy. Similarly, let $E_i(\bar{p})$ denote the first-order expectation about the average price level, and $E_i[\bar{E}[\bar{p}]]$ the second-order expectation about the price level. We can iterate these expectations to the kth higher-order recursively: $\bar{E}^k[X] = \int_0^1 E_j(\bar{E}^{k-1}[X]) dj$.

To characterize how individuals form (higher-order) expectations about the fundamental, we follow Coibion et al. (2021) and assume that individuals do not possess full information. Instead of observing *m* perfectly, they receive one noisy public signal and one private signal. Each signal individually reflects the true value of *m* combined with some noise. In particular, the public signal takes the form: $y = m + \varepsilon$, where $\varepsilon \sim N(0, \kappa_y^{-1})$ and is common across firms. Moreover, each firm *i* also receives its own private signal about $m: x_i = m + v_i$, with $v_i \sim N(0, \kappa_x^{-1})$, and where κ_x and κ_y denote the precision of each type of signal. In order to obtain an individual expectation of *m*, firms weight their signals according to the relative noise in each:

(3)
$$E_i[m] = \frac{\kappa_y}{\kappa_x + \kappa_y} y + \frac{\kappa_x}{\kappa_x + \kappa_y} x_i = (1 - \delta)y + \delta x_i,$$

where $\delta = \frac{\kappa_x}{\kappa_x + \kappa_y}$. The intuition is straightforward, as the private signal becomes more precise relative to the public signal, the firm places relatively more weight on it in when forming beliefs about the fundamental. Aggregating Equation (3) across firms gives the average expectation about the fundamental in the economy:

(4)
$$\overline{E}[m] = (1-\delta)y + \delta m = \int_0^1 E_j(m)dj$$

Firm *i*'s expectation about the average expectation of other managers in the economy is:

(5)
$$E_i[\bar{E}[m]] = (1-\delta)y + \delta E_i[m] = (1-\delta^2)y + \delta^2 x_i$$

By continuing to substitute $E_i[m]$ for m, one can obtain progressively higher-order expectations of m to find:

(6)
$$E_i[\bar{E}^k[m]] = (1 - \delta^{k-1})y + \delta^{k-1}E_i[\bar{E}^{k-1}[m]] = (1 - \delta^k)y + \delta^k x_i$$

Similarly, let $E_i[\bar{p}]$ be the first-order (own) expectation about the price level, and $E_i[\bar{E}[\bar{p}]]$ the higher-order expectation about the price level (i.e. beliefs regarding other managers' beliefs). The average (own) expectation about the price level can be written as:

(7)
$$\bar{E}[\bar{p}] = \omega \bar{\mu} + (1-\omega)\{(1-\alpha)\bar{E}^2[m] + \alpha \bar{E}^2[\bar{p}]\},\$$

Following Coibion et al. (2021), we assume that only first-order expectations are skewed directly. As such, a managers' higher-order expectation is:

(8)
$$E_{i}[\bar{E}[\bar{p}]] = \omega E_{i}[\bar{\mu}] + (1-\omega)\{(1-\alpha)E_{i}[\bar{E}^{2}[m]] + \alpha E_{i}[\bar{E}^{2}[\bar{p}]]\},$$

Continuing this logic, we can identify further higher-order expectations:

(9)
$$\bar{E}^{2}[\bar{p}] = \omega \bar{E}[\bar{\mu}] + (1-\omega)\{(1-\alpha)\bar{E}^{3}[m] + \alpha \bar{E}^{3}[\bar{p}]\},$$

(10)
$$E_i[\bar{E}^2[\bar{p}]] = \omega E_i[\bar{E}[\bar{\mu}]] + (1-\omega)\{(1-\alpha)E_i[\bar{E}^3[m]] + \alpha E_i[\bar{E}^3[\bar{p}]]\},$$

By repeated substitutions in Equation (1), the aggregate price level becomes an average of progressively higher-order expectations of the fundamental, weighted by the complementarities present at each step:

(11)

$$p_{i} = (1 - \alpha)E_{i}[m] + \alpha\omega\mu_{i} + \alpha(1 - \omega)(1 - \alpha)E_{i}[\bar{E}[m]] + \alpha^{2}(1 - \omega)E_{i}[\bar{E}[p]]$$

$$= (1 - \alpha)E_{i}[m] + \alpha\omega\mu_{i} + \alpha(1 - \omega)(1 - \alpha)E_{i}[\bar{E}[m]] + \alpha^{2}(1 - \omega)\omega E_{i}[\bar{\mu}] + \alpha^{2}(1 - \omega)\omega E_{i}[\bar{\mu}] + \alpha^{3}(1 - \omega)^{2}\omega E_{i}[\bar{E}[\bar{\mu}]] + \alpha^{3}(1 - \omega)^{2}\omega E_{i}[\bar{E}^{3}[m]] + \cdots,$$

which can be rewritten as:

(12)
$$p_{i} = \alpha \omega \mu_{i} + \alpha \omega \sum_{k=0}^{\infty} \alpha^{k+1} (1-\omega)^{k+1} E_{i} \Big[\bar{E}^{k} [\bar{\mu}] \Big] + (1$$
$$-\alpha) \sum_{k=0}^{\infty} \alpha^{k} (1-\omega)^{k} [(1-\delta^{k+1})y + \delta^{k+1} x_{i}]$$

Equations (11) and (12) reflect that optimal decisions of firms depend not just on their expectations of the fundamental, but also what they think others think about the fundamental, and so on.

Following Coibion et al. (2021), we next impose some structure on $E_i[\bar{E}^k[\bar{\mu}]]$. Since the optimal price depends on the individual's expectations of the average prior, $\bar{\mu}$, we allow this mean to be unknown, but let each manager observe a private signal of the mean: $\zeta_i \sim N(\bar{\mu}, \kappa_{\zeta}^{-1})$. We assume that the manager's own "long-run" prior skews her view of the aggregate prior:

(13)
$$E_i[\bar{\mu}] = w'\mu_i + (1 - w')\zeta_i$$

where $w' = \frac{\kappa_{\zeta}^{-1}}{(\gamma')^2 + \kappa_{\zeta}^{-1}}$ and $(\gamma')^2 \ge 0$ is again a parameter measuring the degree to which a firm manager prefers her own "long-run" prior when forming beliefs about $\bar{\mu}$. Given that the average of expectations $E_i[\bar{\mu}]$ in Equation (13) is $\bar{E}[\bar{\mu}] = \bar{\mu}$, it follows that $E_i[\bar{E}[\bar{\mu}]] = w'\mu_i + (1 - w')\zeta_i$ and $\bar{E}^2[\bar{\mu}] = \bar{\mu}$. Continuing this logic and by using repeated substitutions, we can show that the expectation for all orders of expectations of the aggregate prior are the same, i.e. $\bar{E}^k[\bar{\mu}] = \bar{\mu}$ for any *k*. Replacing $\bar{E}^k[\bar{\mu}] = \bar{\mu}$ in the optimal pricing equation (12), the formula can be rewritten as:

(14)

$$p_{i} = \alpha \omega \mu_{i} + \alpha \omega \sum_{k=0}^{\infty} \alpha^{k+1} (1-\omega)^{k+1} (w'\mu_{i} + (1-w')\zeta_{i}) + (1-\omega)^{k} (1-\omega)^{k} (1-\omega)^{k} (1-\omega)^{k} (1-\omega)^{k+1} (1-\omega)^{k+1} x_{i})$$

Coibion et al. (2021) proceeds to rewrite Equation (13) in terms of strategies:

(15)
$$p_i = \phi_\mu \mu_i + \phi_\zeta \zeta_i + \phi_x x_i + \phi_y y$$

where

(16)
$$\phi_{\mu} = \alpha \omega \left[1 + \frac{\alpha (1 - \omega)}{1 - \alpha (1 - \omega)} \omega' \right]$$

(17)
$$\phi_{\zeta} = \alpha \omega \left[\frac{\alpha (1-\omega)}{1-\alpha (1-\omega)} (1-\omega') \right]$$

(18)
$$\phi_x = \frac{\delta(1-\alpha)}{1-\alpha\delta(1-\omega)}$$

(19)
$$\phi_y = \frac{(1-\alpha)}{1-\alpha(1-\omega)} - \frac{\delta(1-\alpha)}{1-\alpha\delta(1-\omega)}$$

We can simplify notation by defining

(20)
$$\theta \stackrel{\text{\tiny def}}{=} \phi_{\mu} + \phi_{\zeta} = \frac{\alpha \omega}{1 - \alpha (1 - \omega)}$$

such that

(21)
$$1-\theta = \phi_x + \phi_y = \frac{(1-\alpha)}{1-\alpha(1-\omega)}$$

If long-run priors do not matter, we would receive $\theta = 0$, as both $\omega = \omega' = 0$. Given Equation (15) and substituting for θ , the aggregate price level is:

(22) $\bar{p} = \theta \bar{\mu} + \phi_x m + \phi_y y.$

Expectations about the aggregate price level are formed in line with Equation (22), with the weight assigned to the "long-run" prior:

(23)

$$E_{i}(\bar{p}) = \omega\mu_{i} + (1-\omega)[\theta E_{i}[\bar{\mu}] + (1-\theta)E_{i}(m)]$$

$$= \omega\mu_{i} + (1-\omega)[\theta E_{i}[\bar{\mu}] + \phi_{x}\delta x_{i} + ((1-\phi_{x}\delta) - \theta)y]$$

$$= (\omega + (1-\omega)\theta w')\mu_{i} + (1-\omega)\theta(1-\omega')\zeta_{i} + (1-\omega)[\phi_{x}\delta x_{i} + ((1-\phi_{x}\delta) - \theta)y]$$

The average expected price is thus:

(24)
$$\overline{E}[\overline{p}] = (\omega + (1 - \omega)\theta)\overline{\mu}$$

+
$$(1-\omega) [\phi_x \delta E_i[m] + ((1-\phi_x \delta) - \theta)y]$$

Manager *i* then beliefs other managers to believe:

(25)
$$E_{i}[\bar{E}[\bar{p}]] = (\omega + (1 - \omega)\theta)E_{i}[\bar{\mu}] + (1 - \omega)[\phi_{x}\delta E_{i}[m] + ((1 - \phi_{x}\delta) - \theta)y] = (\omega + (1 - \omega)\theta)\omega'\mu_{i} + (w + (1 - \omega)\theta)(1 - w')\zeta_{i} + (1 - \omega)[\phi_{x}\delta^{2}x_{i} + ((1 - \phi_{x}\delta^{2}) - \theta)y]$$

Average higher-order expectations are thus:

(26)
$$\overline{E}^{2}[\overline{p}] = (\omega + (1 - \omega)\theta)\overline{\mu} + (1 - \omega)[\phi_{x}\delta^{2}m + ((1 - \phi_{x}\delta^{2}) - \theta)y]$$

The difference between average higher-order and first-order expectations is given by:

(27)
$$\overline{E}^{2}[\overline{p}] - \overline{E}[\overline{p}] = (1-\omega)\phi_{x}\delta(1-\delta)(y-m) > 0$$

We follow Coibion et al. (2021) by using Equations (23) - (26) to find expressions for crosssectional disagreement and for forecast uncertainty:

(28)
$$Var[E_i[\bar{p}]] = (\omega + (1-\omega)\theta\omega')^2\kappa_{\mu}^{-1} + ((1-\omega)\theta(1-\omega'))^2\kappa_{\zeta}^{-1} + (1-\omega)^2(\phi_x\delta)^2\kappa_x^{-1}$$

(29)
$$Var\left[E_i[\bar{E}[\bar{p}]]\right]$$

$$= ((\omega + (1 - \omega)\theta)\omega')^2 \kappa_{\mu}^{-1}$$
$$+ ((\omega + (1 - \omega)\theta)(1 - \omega'))^2 \kappa_{\zeta}^{-1}$$
$$+ (1 - \omega)^2 (\phi_x \delta^2)^2 \kappa_x^{-1}$$

(30)
$$\Omega_{\{E_i[\bar{p}]|y\}} = \left((1-\omega)\theta(1-\omega')\right)^2 \kappa_{\zeta}^{-1} + (1-\omega)^2 (\phi_x \delta)^2 \kappa_x^{-1}$$

(31)
$$\Omega_{\{E_i[\bar{E}[\bar{p}]]|y\}} = ((\omega + (1 - \omega)\theta)(1 - \omega'))^2 \kappa_{\zeta}^{-1} + (1 - \omega)^2 (\phi_x \delta^2)^2 \kappa_x^{-1}$$

The derivations in Equations (30) and (31) assume that each manager knows his own "long-run" prior with certainty and is not considering that his "long-run" prior differs from the aggregate prior. Importantly, by using Equations (28) and (29), we can investigate the relative magnitude of disagreement in first-order and higher-order expectations. In a similar spirit, we can use Equations (30) and (31) to investigate the relative magnitude of uncertainty in first-order and higher-order expectations. We start by deriving predictions regarding disagreement in first- and higher-order expectations:

(32)

$$Var\left[E_{i}\left[\bar{E}[\bar{p}]\right]\right] - Var\left[E_{i}[\bar{p}]\right]$$

$$= \left(2(1-\omega)(\omega\theta\omega')(\omega'-1) + \omega^{2}(\omega'^{2}-2)\right)\kappa_{\mu}^{-1}$$

$$+ \left[\omega\left((\omega+(1-\omega)\theta) + \left((1-\omega)\theta\right)\right)\right](1)$$

$$- \omega')^{2}\kappa_{\zeta}^{-1} + (1-\omega)^{2}(\phi_{x}\delta)^{2}(\delta^{2}-1)\kappa_{x}^{-1}$$

Next, we derive predictions regarding uncertainty in first- and higher-order expectations:

(33)

$$\Omega_{\{E_i[\bar{E}[\bar{p}]]|y\}} - \Omega_{\{E_i[\bar{p}]|y\}}$$

$$= \left[\omega\left((\omega + (1-\omega)\theta) + ((1-\omega)\theta)\right)\right](1-\omega')^2\kappa_{\zeta}^{-1} + (1-\omega)^2(\phi_x\delta)^2(\delta^2-1)\kappa_x^{-1}$$

Importantly, the expressions for disagreement and uncertainty in Equations (32) and (33) do not make a clear prediction regarding the relative magnitude of uncertainty in first- and higher-order expectations. In contrast, in the baseline model of Coibion et al. (2021), uncertainty (disagreement) in higher-order expectations is always lower than uncertainty (disagreement) in first-order expectations. In the extension with heterogenous "long-run" priors, this relation is ambiguous and depends on relative magnitudes of κ_{ζ}^{-1} and κ_{x}^{-1} . In particular, if κ_{x}^{-1} and κ_{μ}^{-1} are large relative to κ_{ζ}^{-1} the model produces $\Omega_{\{E_{i}[\bar{E}[\bar{p}]]|y\}} < \Omega_{\{E_{i}[\bar{p}]|y\}}$ as well as $Var\left[E_{i}[\bar{E}[\bar{p}]]\right] < Var\left[E_{i}[\bar{p}]\right]$, which is the finding of Coibion et al. (2021). However, if κ_{x}^{-1} and κ_{μ}^{-1} are small relative to κ_{ζ}^{-1} , the model produces $\Omega_{\{E_{i}[\bar{E}[\bar{p}]]|y\}} > \Omega_{\{E_{i}[\bar{p}]|y\}}$ as well as $Var\left[E_{i}[\bar{E}[\bar{p}]]\right] > Var\left[E_{i}[\bar{p}]\right]$, which is consistent with our findings. It is important to note that the relation between uncertainty and disagreement in first- versus higher-order expectations always moves in tandem as they rely on the same conditions. University of Innsbruck - Working Papers in Economics and Statistics Recent Papers can be accessed on the following webpage:

https://www.uibk.ac.at/eeecon/wopec/

- 2023-10 Pascal Kieren, Christian König-Kersting, Robert Schmidt, Stefan Trautmann, Franziska Heinicke: First-Order and Higher-Order Inflation Expectations: Evidence about Households and Firms
- 2023-09 Marica Valente, Timm Gries, Lorenzo Trapani: Informal employment from migration shocks
- 2023-08 Natalie Struwe, Esther Blanco, James M. Walker: No response to changes in marginal incentives in one-shot public good experiments
- 2023-07 Sebastian Bachler, Sarah Lynn Flecke, Jürgen Huber, Michael Kirchler, Rene Schwaiger: Carbon Pricing, Carbon Dividends and Cooperation: Experimental Evidence
- 2023-06 Elisabeth Gsottbauer, Michael Kirchler, and Christian König-Kersting: Climate Crisis Attitudes among Financial Professionals and Climate Experts
- 2023-05 Xiaogeng Xu, Satu Metsälampi, Michael Kirchler, Kaisa Kotakorpi, Peter Hans Matthews, Topi Miettinen: Which income comparisons matter to people, and how? Evidence from a large field experiment
- 2023-04 **Tanja Hoertnagl, Rudolf Kerschbamer, Regine Oexl, Rudi Stracke, and Uwe Sunde:** Heterogeneity in Rent-Seeking Contests with Multiple Stages: Theory and Experimental Evidence
- 2023-03 Melissa Newham, Marica Valente: The Cost of Influence: How Gifts to Physicians Shape Prescriptions and Drug Costs
- 2023-02 Natalie Struwe, Esther Blanco, James M. Walker: Determinants of Financial Literacy and Behavioral Bias among Adolescents
- 2023-01 Marco Aschenwald, Armando Holzknecht, Michael Kirchler, Michael Razen: Determinants of Financial Literacy and Behavioral Bias among Adolescents
- 2022-20 Silvia Angerer, Daniela Glätzle-Rützler, Philipp Lergetporer, and Thomas Rittmannsberger: Beliefs about social norms and (the polarization of) COVID-19 vaccination readiness
- 2022-19 Edward I. Altman, Marco Balzano, Alessandro Giannozzi, Stjepan Srhoj: Revisiting SME default predictors: The Omega Score
- 2022-18 Johannes Diederich, Raphael Epperson, Timo Goeschl: How to Design the Ask? Funding Units vs. Giving Money

- 2022-17 **Toman Barsbai, Vojtěch Bartoš, Victoria Licuanan, Andreas Steinmayr, Erwin Tiongson, and Dean Yang:** Picture This: Social Distance and the Mistreatment of Migrant Workers
- 2022-16 Andreas Steinmayr, Manuel Rossi: Vaccine-skeptic physicians and COVID-19 vaccination rates
- 2022-15 Stjepan Srhoj, Alex Coad, Janette Walde: HGX: The Anatomy of High Growth Exporters
- 2022-14 Martin Obradovits, Philipp Plaickner Price-Directed Search, Product Differentiation and Competition
- 2022-13 Utz Weitzel, Michael Kirchler The Banker's Oath And Financial Advice
- 2022-12 Julian Granna, Wolfgan Brunauer, Stefan Lang: Proposing a global model to manage the bias-variance tradeoff in the context of hedonic house price models
- 2022-11 Christoph Baumgartner, Stjepan Srhoj and Janette Walde: Harmonization of product classifications: A consistent time series of economic trade activities
- 2022-10 Katharina Momsen, Markus Ohndorf: Seller Opportunism in Credence Good Markets ? The Role of Market Conditions
- 2022-09 Christoph Huber, Michael Kirchler: Experiments in Finance ? A Survey of Historical Trends
- 2022-08 Tri Vi Dang, Xiaoxi Liu, Florian Morath: Taxation, Information Acquisition, and Trade in Decentralized Markets: Theory and Test
- 2022-07 Christoph Huber, Christian König-Kersting: Experimenting with Financial Professionals
- 2022-06 Martin Gächter, Martin Geiger, Elias Hasler: On the structural determinants of growth-at-risk
- 2022-05 Katharina Momsen, Sebastian O. Schneider: Motivated Reasoning, Information Avoidance, and Default Bias
- 2022-04 Silvia Angerer, Daniela Glätzle-Rützler, Philipp Lergetporer, Thomas Rittmannsberger: How does the vaccine approval procedure affect COVID-19 vaccination intentions?
- 2022-03 Robert Böhm, Cornelia Betsch, Yana Litovsky, Philipp Sprengholz, Noel Brewer, Gretchen Chapman, Julie Leask, George Loewenstein, Martha Scherzer, Cass R. Sunstein, Michael Kirchler: Crowdsourcing interventions to promote uptake of COVID-19 booster vaccines

- 2022-02 Matthias Stefan, Martin Holmén, Felix Holzmeister, Michael Kirchler, Erik Wengström: You can't always get what you want-An experiment on finance professionals' decisions for others
- 2022-01 **Toman Barsbai, Andreas Steinmayr, Christoph Winter:** Immigrating into a Recession: Evidence from Family Migrants to the U.S.
- 2021-32 Fanny Dellinger: Housing Support Policies and Refugees' Labor Market Integration in Austria
- 2021-31 Albert J. Menkveld, Anna Dreber, Felix Holzmeister, Jürgen Huber, Magnus Johannesson, Michael Kirchler, Sebastian Neusüss, Michael Razen, Utz Weitzel and et al: Non-Standard Errors
- 2021-30 Toman Barsbai, Victoria Licuanan, Andreas Steinmayr, Erwin Tiongson, Dean Yang: Information and Immigrant Settlement
- 2021-29 Natalie Struwe, Esther Blanco, James M. Walker: Competition Among Public Good Providers for Donor Rewards
- 2021-28 **Stjepan Srhoj, Melko Dragojević:** Public procurement and supplier job creation: Insights from auctions
- 2021-27 **Rudolf Kerschbamer, Regine Oexl:** The effect of random shocks on reciprocal behavior in dynamic principal-agent settings
- 2021-26 Glenn E. Dutcher, Regine Oexl, Dmitry Ryvkin, Tim Salmon: Competitive versus cooperative incentives in team production with heterogeneous agents
- 2021-25 Anita Gantner, Regine Oexl: Respecting Entitlements in Legislative Bargaining A Matter of Preference or Necessity?
- 2021-24 Silvia Angerer, E. Glenn Dutcher, Daniela Glätzle-Rützler, Philipp Lergetporer, Matthias Sutter: The formation of risk preferences throughsmall-scale events
- 2021-23 Stjepan Srhoj, Dejan Kovač, Jacob N. Shapiro, Randall K. Filer: The Impact of Delay: Evidence from Formal Out-of-Court Restructuring
- 2021-22 Octavio Fernández-Amador, Joseph F. Francois, Doris A. Oberdabernig, Patrick Tomberger: Energy footprints and the international trade network: A new dataset. Is the European Union doing it better?
- 2021-21 Felix Holzmeister, Jürgen Huber, Michael Kirchler, Rene Schwaiger: Nudging Debtors to Pay Their Debt: Two Randomized Controlled Trials
- 2021-20 Daniel Müller, Elisabeth Gsottbauer: Why Do People Demand Rent Control?
- 2021-19 Alexandra Baier, Loukas Balafoutas, Tarek Jaber-Lopez: Ostracism and Theft in Heterogeneous Groups

- 2021-18 Zvonimir Bašić, Parampreet C. Bindra, Daniela Glätzle-Rützler, Angelo Romano, Matthias Sutter, Claudia Zoller: The roots of cooperation
- 2021-17 Silvia Angerer, Jana Bolvashenkova, Daniela Glätzle-Rützler, Philipp Lergetporer, Matthias Sutter: Children's patience and school-track choices several years later: Linking experimental and field data
- 2021-16 **Daniel Gründler, Eric Mayer, Johann Scharler:** Monetary Policy Announcements, Information Schocks, and Exchange Rate Dynamics
- 2021-15 Sebastian Bachler, Felix Holzmeister, Michael Razen, Matthias Stefan: The Impact of Presentation Format and Choice Architecture on Portfolio Allocations: Experimental Evidence
- 2021-14 Jeppe Christoffersen, Felix Holzmeister, Thomas Plenborg: What is Risk to Managers?
- 2021-13 Silvia Angerer, Daniela Glätzle-Rützler, Christian Waibel: Trust in health care credence goods: Experimental evidence on framing and subject pool effects
- 2021-12 Rene Schwaiger, Laura Hueber: Do MTurkers Exhibit Myopic Loss Aversion?
- 2021-11 Felix Holzmeister, Christoph Huber, Stefan Palan: A Critical Perspective on the Conceptualization of Risk in Behavioral and Experimental Finance
- 2021-10 Michael Razen, Alexander Kupfer: Can increased tax transparency curb corporate tax avoidance?
- 2021-09 Changxia Ke, Florian Morath, Anthony Newell, Lionel Page: Too big to prevail: The paradox of power in coalition formation
- 2021-08 Marco Haan, Pim Heijnen, Martin Obradovits: Competition with List Prices
- 2021-07 Martin Dufwenberg, Olof Johansson-Stenman, Michael Kirchler, Florian Lindner, Rene Schwaiger: Mean Markets or Kind Commerce?
- 2021-06 Christoph Huber, Jürgen Huber, and Michael Kirchler: Volatility Shocks and Investment Behavior
- 2021-05 Max Breitenlechner, Georgios Georgiadis, Ben Schumann: What goes around comes around: How large are spillbacks from US monetary policy?
- 2021-04 Utz Weitzel, Michael Kirchler: The Banker's Oath And Financial Advice
- 2021-03 Martin Holmen, Felix Holzmeister, Michael Kirchler, Matthias Stefan, Erik Wengström: Economic Preferences and Personality Traits Among Finance Professionals and the General Population
- 2021-02 Christian König-Kersting: On the Robustness of Social Norm Elicitation

2021-01 Laura Hueber, Rene Schwaiger: Debiasing Through Experience Sampling: The Case of Myopic Loss Aversion.

University of Innsbruck

Working Papers in Economics and Statistics

2023-10

Pascal Kieren, Christian König-Kersting, Robert Schmidt, Stefan Trautmann, Franziska Heinicke

First-Order and Higher-Order Inflation Expectations: Evidence about Households and Firms

Abstract

We study first-order and higher-order inflation expectations of German households and firms elicited from surveys. The data allows to shed light on the relation between different orders of beliefs, and to derivate implications for noisy-information models with infinite regress. Moreover, since the elicited data is identical for households and firms, it also allows studying whether the relation between first-order and higher-order beliefs differs between the two samples. While we find that this relation is mostly identical between households and firms in our data, we identify differences to previously elicited data in the literature. We discuss potential sources for these differences and their theoretical implications.

ISSN 1993-4378 (Print) ISSN 1993-6885 (Online)